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Abstract 

The thesis focuses on the study of space-/time-resolved phase volume fraction distribution 

of unary and binary gas-solid flows in a cylindrical fluidized bed. The thesis particularly 

deliberates on the characterization of time-/space-resolved phase distribution in unary and 

binary gas-solids flow and also segregation through measurements using the Electrical 

Capacitance Tomography (ECT), and through CFD simulations using two-/multi-fluid 

Eulerian models.  

The ability of two-fluid Eulerian model to predict the instantaneous phase distribution was 

studied. The measured time-averaged phase distributions, bubble chord length distributions and 

bubbling frequency are compared with the corresponding predictions obtained from two-fluid 

Eulerian simulations for unary gas-solid flows. The results showed that the predicted time-

averaged solid volume fraction distribution differed marginally and was in a satisfactory 

agreement with the corresponding measurements. However, the bubbling frequency and bubble 

chord length distribution predicted using different drag models, in particular, the formation of 

large bubbles/slugs, were considerably different. A modified drag model is proposed that led 

to a reduction in the regions in the bed in which higher drag force was applied, allowed the 

formation of larger bubbles/slugs and led to an improvement in the predicted bubbling 

behavior. It was also found that the use of different drag models, in turn, also affects the total 

solid viscosity and therefore the bubbling behavior. 

The posterior analysis showed that among solid-phase kinematic, collisional, and 

frictional viscosities, the influence of frictional viscosity on the bubbling behavior was found 

to be more dominant. Also, the magnitude of frictional viscosity often goes up to more than 

1000 times of the collisional and kinetic viscosity in the fluid domain. An increase in the 

frictional viscosity leads to compaction of the solid phase and increase in the formation of 
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large bubbles/slugs, those were observed experimentally. The predicted bubbling frequency 

and bubble chord length distribution were in a quantitative agreement with those measured 

using the ECT for different gas velocities and particle sizes.  

Further, the ECT technique was applied to measure the segregation time for binary gas-

solids flow in a cylindrical fluidized bed. The measurements of segregation and fluidization of 

binary mixtures were investigated using the ECT and high-speed imaging. It was shown that 

ECT can be used to detect the regions of complete segregation in a cylindrical fluidized beds. 

In particular, the binary gas-solids flows in systems with large D/dp ratio, which are otherwise 

difficult to characterize using high-speed imaging. Under the fluidization conditions, the 

bubbling behavior of the mixtures was governed by the smaller-sized particles in spite of 

having an equal percentage by weight of the particles in the mixture. For the fluidization of 

binary mixtures, a relation between the measured variance of solid volume fraction fluctuations 

and the corresponding flow structures was studied. It was also found that different bubbling 

behaviors can produce the same time-averaged solid volume fraction profiles. As a 

consequence, different phase distributions can lead to different heat and mass transport rates 

depending on the local interfacial area, despite having the same time-averaged phase hold-up 

distribution. 

Further, efforts are made to predict the segregation time using the Eulerian multi-fluid 

model. It was seen that due to higher gas-particle drag force magnitude, the larger particles 

were carried away from bottom to the top of the bed resulting into less extent of segregation 

and higher segregation time in comparison to the corresponding experiments. The collisional 

contribution in the particle-particle drag model, led to increase in the segregation time, and 

therefore it was neglected and frictional coefficient of 0.5 was used which improved the 

agreement with the measured segregation time. In fluidized bed conditions for binary mixtures, 
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the predicted distribution was in a qualitative agreement. The reason for quantitative difference 

was due to ad-hoc values of different parameters used in particle-particle drag models for 

collision and friction interactions between the solids. 

The present thesis helps to understand the characteristics of instantaneous phase 

distribution and segregation in gas-solids fluidization. It also helps in accessing the ability of 

Eulerian multi-fluid CFD models to predict the dynamics of instantaneous phase distribution 

(in terms of bubble size distribution and bubbling frequency) and segregation in gas-solids 

fluidization.  

 

   



 

सार 
थीससस अतंररक्ष के अध्ययन पर कें द्रित है - एक बेलनाकार िवीकृत बबस्तर में एकता और बाइनरी 
गैस-ठोस प्रवाह के अतंररक्ष-समय-ननर्ााररत चरण मात्रा अशं ववतरण। थीससस ववशषे रूप से असमान 
और द्ववआर्ारी गैस-ठोस प्रवाह में समय अतंररक्ष-समार्ान चरण ववतरण के ववचलन पर ववचार-
ववमशा करता है और ववद्युत संर्ाररत्र टोमोग्राफी (ईसीटी) का उपयोग करके माप के माध्यम से 
अलगाव, और दो- / बहु-िव यूलररयन का उपयोग करके सीएफडी ससमुलेशन के माध्यम से करता 
है। मॉडल। 
तात्कासलक चरण ववतरण की भववष्यवाणी करने के सलए दो-िव यूलररयन मॉडल की क्षमता का 
अध्ययन ककया गया था। मापा समय-औसत चरण ववतरण, बुलबुला कॉडा लंबाई ववतरण और 
बुदबुदाहट आवतृ्त्त की तुलना गैस-ठोस प्रवाह के सलए दो-िव यूलर ससमुलेशन से प्राप्त संगत 
भववष्यवाणणयों के साथ की जाती है। पररणामों से पता चलता है कक अनुमाननत समय-औसत ठोस 
मात्रा अशं ववतरण सभन्न रूप से सभन्न था और इसी माप के साथ एक संतोषजनक समझौत ेमें 
था। हालांकक, बुदबुदाहट आवतृ्त्त और बबल कॉडा लंबाई ववतरण ने अलग-अलग ड्रगै मॉडल का 
उपयोग करके भववष्यवाणी की, ववशषे रूप से, बड ेबुलबुले / स्लग के गठन, काफी सभन्न थे। एक 
संशोधर्त ड्रगै मॉडल प्रस्ताववत ककया गया है त्जससे बेड में उन क्षेत्रों में कमी आई है त्जसमें उच्च 
ड्रगै फोसा लागू ककया गया था, त्जससे बड ेबुलबुले / स्लग के गठन की अनुमनत समली और 
पूवाानुमाननत बुदबुदाहट व्यवहार में सुर्ार हुआ। यह भी पाया गया कक अलग-अलग ड्रगै मॉडल का 
उपयोग, बदले में, कुल ठोस धचपधचपाहट और इससलए बुदबुदाने वाले व्यवहार को भी प्रभाववत 
करता है। 

पीछे के विश्लेषण से पता चला है कक ठोस-चरण कीनेमेटीक्स, कोलाइज़ल और घषषण चचपचचपाहट 
के बीच, बुदबुदाती हुई व्यिहार पर घषषण चचपचचपाहट का प्रभाि अचिक प्रभािी पाया गया। इसके 
अलािा, घषषण चचपचचपाहट की भयािहता अक्सर द्रि क्षेत्र में टकराि और गततज चचपचचपाहट के 
1000 स ेअचिक बार तक जाती है। घषषण चचपचचपाहट में िवृि से ठोस चरण का संघनन होता है 
और बड ेबुलबुले / स्लग के तनमाषण में िवृि होती है, जजन्हें प्रायोचगक तौर पर देखा गया था। 
पूिाषनुमातनत बुदबुदाहट आिजृतत और बबल कॉर्ष लंबाई वितरण विभभन्न गैस िेगों और कण आकारों 
के भलए ईसीटी का उपयोग करके मापा गया उन लोगों के साथ एक मात्रातमक समझौते में थे। 

इसके अलािा, बेलनाकार द्रवित बबस्तर में द्विआिारी गैस-ठोस प्रिाह के भलए अलगाि समय को 
मापने के भलए ईसीटी तकनीक लागू की गई थी। ईसीटी और उच्च गतत इमेजजंग का उपयोग 
करके द्विआिारी भमश्रण के अलगाि और द्रिीकरण के माप की जांच की गई। यह ददखाया गया 



 

कक ईसीटी का उपयोग एक बेलनाकार द्रिीकृत बेर् में पूणष अलगाि के क्षेत्रों का पता लगाने के 
भलए ककया जा सकता है। विशषे रूप से, बाइनरी गसै-सोभलड्स बड ेर्ी / र्ीपी अनुपात िाले भसस्टम 
में बहते हैं, जो उच्च गतत इमेजजंग का उपयोग करके चचजननत करना मुजश्कल है। विखरं्न की 
जस्थतत के तहत, भमश्रण के बबभलगं व्यिहार को भमश्रण में कणों के िजन के बराबर प्रततशत होन े
के बािजूद छोटे आकार के कणों द्िारा तनयंबत्रत ककया गया था। बाइनरी भमश्रण के द्रिीकरण के 
भलए, ठोस मात्रा अशं में उतार-चढाि और संबंचित प्रिाह संरचनाओं के मापा विचरण के बीच एक 
संबंि का अध्ययन ककया गया था। यह भी पाया गया कक विभभन्न बुदबुदाहट व्यिहार एक ही 
समय-औसत ठोस मात्रा अशं प्रोफाइल का उतपादन कर सकते हैं। एक पररणाम के रूप में, अलग-
अलग चरण के वितरण में एक ही समय-औसत चरण होल्र्-अप वितरण होने के बािजूद, स्थानीय 
इंटरफेभशयल क्षेत्र के आिार पर विभभन्न गमी और बड ेपैमान ेपर पररिहन दरों का कारण बन 
सकता है। 

इसके अलावा, यूलररयन बहु-िव मॉडल का उपयोग करके अलगाव के समय की भववष्यवाणी करने 
का प्रयास ककया जाता है। यह देखा गया कक उच्च गैस-कण ड्रगै फोसा पररमाण के कारण, बड े
कणों को बेड के ऊपर से नीचे तक ले जाया गया था, त्जसके पररणामस्वरूप संबंधर्त प्रयोगों की 
तुलना में अलगाव और कम अलगाव समय था। कण-कण ड्रगै मॉडल में समवती योगदान, अलगाव 
के समय में ववृि का कारण बना, और इससलए इसे उपेक्षक्षत ककया गया और 0.5 के घषाण गुणांक 
का उपयोग ककया गया, त्जसने मापा पथृक्करण समय के साथ समझौते में सुर्ार ककया। बाइनरी 
समश्रण के सलए िववत बबस्तर की त्स्थनत में, अनुमाननत ववतरण एक गुणात्मक समझौते में था। 
मात्रात्मक अतंर का कारण ठोस पदाथों के बीच टकराव और घषाण इंटरैक्शन के सलए कण-कण 
ड्रगै मॉडल में उपयोग ककए जाने वाले ववसभन्न मापदंडों के तदथा मूल्यों के कारण था। 

वतामान थीससस गैस-ठोस तरल पदाथा में तात्कासलक चरण ववतरण और अलगाव की ववशषेताओं 
को समझने में मदद करता है। यह तात्कासलक चरण ववतरण (बुलबुला आकार ववतरण और 
बुदबुदाहट आवतृ्त्त के संदभा में) और गैस-ठोस िव के अलगाव में अलगाव की गनतशीलता की 
भववष्यवाणी करन ेके सलए यूसलयन मल्टी-िव सीएफडी मॉडल की क्षमता तक पहंुचने में भी मदद 
करता है। 
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Nomenclature 
𝐶𝐷  drag coefficient, - 

𝐶𝑓𝑟  friction coefficient, - 

𝐷𝑏 fluidized bed (internal) diameter, cm 

𝑑𝑠 solid particle diameter, μm   

𝑑ℎ horizontal bubble chord length, cm   

𝑑𝑣 vertical bubble chord, s   

𝑒𝑠𝑠 particle-particle restitution co–efficient, - 

𝑔 gravitational acceleration, m/s2 

𝑔0,𝑠𝑠 radial distribution function, - 

𝐻𝑏 bed height, cm 

𝐻𝑑 heterogeneity index, - 

𝐻𝑠 static bed height, cm 

𝐼2𝐷 second invariant of the strain rate tensor, 1/s2 

𝐼 identity tensor, - 

𝐾𝑔𝑠 gas-solid interphase momentum exchange coefficient, kg/(m3∙s) 

𝐾𝑠𝑠, 𝐾𝑖𝑗 solid-solid interphase momentum exchange coefficient, kg/(m3∙s) 

li chord length, cm 

l32 Sauter mean chord length, cm 

𝑝𝑠 solid pressure, Pa 

𝑃𝑠,𝑓 solid frictional pressure, Pa 

𝑅𝑒𝑠 particles Reynolds number, - 

𝑡 flow time, s 

𝑈 velocity, m/s 

𝑈𝐺 superficial gas velocity, m/s 

  

Greek Letters 

α instantaneous phase volume fraction, - 

𝛼𝑖 gas volume fraction value for dividing criteria in the switch function, - 

𝛼𝑠𝑡 total solids volume fraction, - 

𝛼̅𝑠 time-averaged solid volume fraction, - 
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𝛼𝑠,𝑚𝑎𝑥 maximum solid packing limit, - 

𝛼𝑠,𝑓,𝑚𝑖𝑛 critical solid volume fraction, - 

𝜇 viscosity, Pa∙s 

ρ density, kg/m3 

𝜏̿ shear stress, N/m2 

𝜀 relative permittivity, - 

𝛽 interphase momentum exchange coefficient, kg/(m3s) 

𝜑 specularity co–efficient, - 

𝜆𝑠 solid bulk viscosity, Pa.s 

𝜅𝜃,𝑠 granular diffusion co-efficient, kg/(m.s) 

𝛾𝜃,𝑠 granular dissipation, W/m3 

  granular temperature, m2/s2 

  switch function, - 

  angle of internal friction, degree 

Subscripts 

col collisional 

fr friction 

𝑔 gas 

kin kinetic 

𝑚 mixture 

𝑚𝑓 minimum fluidization 

𝑟 relative 

𝑠 solid phase 

𝑠𝑓 solid phase friction 

Acronyms 

CFD Computational Fluid Dynamics 

ECT Electrical Capacitance Tomography 

ILBP Iterative Linear Back Projection 

KTGF Kinetic Theory of Granular Flow 

TFM Two-Fluid Model 
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